**Author’s response to reviews**

**Title:** Effect of metastatic site on survival in patients with neuroendocrine neoplasms (NENs). An analysis of SEER data from 2010 -2014

**Authors:**

Nikolaos Trikalinos (ntrikalinos@wustl.edu)

Benjamin Tan (btan@wustl.edu)

Manik Amin (manikamin@wustl.edu)

Jingxia Liu (esther@wustl.edu)

Ramaswamy Govindan (rgovindan@wustl.edu)

Daniel Morgensztern (danielmorgensztern@wustl.edu)

**Version:** 2  **Date:** 26 Feb 2020

**Author’s response to reviews:**

Thank you very much for the suggestions for improvement. We have made the relevant changes.

Regards,
Nikolaos A Trikalinos (on behalf of all authors).

1. Please change the section heading “Introduction” to “Background”. It has been corrected.

2. Please change the section heading “Materials and Methods” to “Methods”. It has been corrected.

3. Please add a “Conclusions” section after the “Discussion”. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance. It has been added.

4. As many abbreviations are used throughout the paper, we ask that you include a list of abbreviations (before the declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use. It has been added.

5. In the 'Availability of data and materials' section of your Declarations please add links/references to the SEER database used in your study. Please state whether public access to the database(s) is open or closed. If closed, please state whether your received administrative permission to access and use these. It has been corrected.
6. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors. It has been corrected.

7. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case. It has been corrected.

8. We note on Line 228 you have a duplicated "Acknowledgements" section. Please remove this duplication. It has been corrected.

9. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. It has been corrected.